

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Lee, Hyun-Hoon; Park, Donghyun; Wang, Jing

Working Paper The Role of the People's Republic of China in International Fragmentation and Production Networks: An Empirical Investigation

ADB Working Paper Series on Regional Economic Integration, No. 87

Provided in Cooperation with: Asian Development Bank (ADB), Manila

Suggested Citation: Lee, Hyun-Hoon; Park, Donghyun; Wang, Jing (2011) : The Role of the People's Republic of China in International Fragmentation and Production Networks: An Empirical Investigation, ADB Working Paper Series on Regional Economic Integration, No. 87, Asian Development Bank (ADB), Manila, https://hdl.handle.net/11540/2008

This Version is available at: https://hdl.handle.net/10419/109590

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

http://creativecommons.org/licenses/by/3.0/igo

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





ADB Working Paper Series on Regional Economic Integration



The Role of the People's Republic of China in International Fragmentation and Production Networks: An Empirical Investigation

Hyun-Hoon Lee, Donghyun Park, and Jing Wang No. 87 | September 2011

Asian Development Bank



ADB Working Paper Series on Regional Economic Integration

The Role of the People's Republic of China in International Fragmentation and Production Networks: An Empirical Investigation

Hyun-Hoon Lee⁺, Donghyun Park⁺⁺, and Jing Wang⁺⁺⁺

No. 87 | September 2011

An earlier version of this paper was presented at the 9th Asia-Pacific Economic Forum Conference held at Renmin University in the People's Republic of China in October 2010. We wish to thank the participants of the conference, as well as Lei Lei Song and Thiam Hee Ng of the Asian Development Bank, for their many useful comments.

*Professor, Department of International Trade, Kangwon National University, Chuncheon 200-701, Republic of Korea. hhlee@kangwon.ac.kr

⁺⁺Principal Economist, Economics and Research Department, Asian Development Bank, 6 ADB Avenue, Mandaluyong City, Metro Manila, Philippines. 1550. dpark@adb.org

***Ph.D. Student, Department of International Trade, Kangwon National University, Chuncheon 200-701, Republic of Korea. wangjing6526@hotmail.com

Asian Development Bank

The ADB Working Paper Series on Regional Economic Integration focuses on topics relating to regional cooperation and integration in the areas of infrastructure and software, trade and investment, money and finance, and regional public goods. The Series is a quick-disseminating, informal publication that seeks to provide information, generate discussion, and elicit comments. Working papers published under this Series may subsequently be published elsewhere.

Disclaimer:

The views expressed in this paper are those of the author and do not necessarily reflect the views and policies of the Asian Development Bank or its Board of Governors or the governments they represent.

The Asian Development Bank does not guarantee the accuracy of the data included in this publication and accepts no responsibility for any consequence of their use.

Use of the term "country" does not imply any judgment by the authors or the Asian Development Bank as to the legal or other status of any territorial entity.

Unless otherwise noted, \$ refers to US dollars.

© 2011 by Asian Development Bank September 2011 Publication Stock No. WPS113969

Contents

Ab	stract		V
1.	Introd	duction	1
2.	Litera	ature Review	2
	2.1.	Production Sharing in the World	2
	2.2.	The People's Republic of China's Role in Production Sharing	3
3.	Struc	ture of the Parts and Components Trade	4
	3.1.	Trends and Patterns of World Trade in Parts and Components	4
	3.2.	Overall Structure of the People's Republic of China's Trade in Parts and Components	5
	3.3.	Structure of the People's Republic of China's Parts and Components Trade	8
	3.4.	The People's Republic of China's Parts and Components Trade—Destination and Source Countries	14
4.	The [in Pa	Determinants of the People's Republic of China's Trade rts and Components	16
	4.1.	The Gravity Equation	16
	4.2.	The Augmented Gravity Equation for the People's Republic of China's Trade in Parts and Components	17
	4.3	Estimation Results	18
5.	Conc	luding Observations	25
Re	ferenc	es	27
AD)B Wo	rking Paper Series on Regional Economic Integration	30

Tables

1a.	The People's Republic of China's Export Trade in Parts and Components	9
1b.	The People's Republic of China's Import Trade in Parts and Components	10
2a.	Percentage of Parts and Components in the People's Republic of China's Exports by 2-Digit SITC Categories	12
2b.	Percentage of Parts and Components in the People's Republic of China's Imports by 2-Digit SITC Categories	13
За.	The People's Republic of China's Parts and Components Export Share by Major Trading Partners and Regions	15
3b.	The People's Republic of China's Parts and Components Import Share by Major Trading Partners and Regions	16
4.	Determinants of the People's Republic of China's Exports —Fixed Effects Model	19
5.	Determinants of the People's Republic of China's Imports —Fixed Effects Model	20
6.	Determinants of the People's Republic of China's Exports — Random Effects Model	23
7.	Determinants of the People's Republic of China's Imports — Random Effects Model	24
Fig	ures	
1.	World Trade in Parts and Components	5
2.	The People's Republic of China's Trade in Parts and Components	6
3.	Parts and Components Trade as a Share of Total Trade —The People's Republic of China and the World Average	7
4.	The People's Republic of China's Exports and Imports —Parts and Components vs. Final Goods	7
5.	Export Destination Countries for the Parts and Components Trade of the People's Republic of China	14

6. Import Source Countries for the Parts and Components Trade of the People's Republic of China15

Abstract

Despite the central role of the People's Republic of China (PRC) in global parts and components trade, most previous studies on the PRC's parts and components trade have been limited to a particular trade partner or some specific industries. The central objective of this paper is to provide a more complete description of the PRC's parts and components trade. To do so, we systematically separate total trade flows into parts and components trade for the period 1992–2009. We then estimate a gravity model to examine the determinants of the PRC's trade in parts and components. We find that the share of parts and components trade in the PRC's total trade has grown rapidly.

Keywords: People's Republic of China, PRC, parts and components trade, fragmentation, production network

JEL Classification: F14, F21, F23

1. Introduction

The stunning rise of the PRC as a globally influential economic power since its opening in 1978 closely parallels its equally impressive rise as a globally significant trading power. Trade has been the primary channel through which the PRC has been transformed from a closed autarkic economy to an open globalized one. Much of the PRC's trade, in turn, is driven by its central role—as the "Factory of the World"—in international production networks that are based on cross-border flows of parts and components culminating in their assembly into final products. Such flows have been particularly prominent in the PRC's trade with its neighbors, and the resulting regional production network has consolidated East Asia's status as the world's manufacturing hub. At a broader level, the central objective of this paper is to provide a comprehensive analysis of the PRC's trade in parts and components. Such an analysis would significantly increase understanding of the PRC's foreign trade and—by extension—its rapid growth and development.

International production fragmentation refers to the cross-border dispersion of components' production and assembly within vertically integrated production processes, with each country specializing in a particular stage of the production sequence. This process has become an important feature of the world economy (Athukorala and Yamashita 2006). In particular, the PRC's rapid growth and development is related to its growing participation in international production networks. The role of the PRC as a globally significant assembly and production center is deepening its integration into the world economy and facilitating the greater fragmentation of production across countries in East Asia.

There have been many studies confirming the existence of well-established East Asian intra-industry production networks. Such studies have also found that the PRC's increasing integration into the world economy has facilitated the greater fragmentation of production across Asian countries. Despite the central role of the PRC in the regional parts and components trade, there has been no comprehensive study on the topic and most previous studies of the PRC's parts and components trade have been limited to a particular trading partner or specific industries. To address this shortcoming, we seek to give a more complete description of the PRC's parts and components trade. More specifically, we aim to

- i. comprehensively document the magnitude of the PRC's participation in international production networks and its changing over time,
- ii. identify the position of the PRC in international production networks and the PRC's major trading partners in the parts and components trade, and
- iii. examine the various factors that affect the PRC's imports and exports of parts and components.

In order to accomplish the above goals, we systematically separate out intermediate goods (parts and components) and final goods from total trade flows, and give a description of the pattern and trends of the parts and components trade of the PRC during 1992–2009. We then estimate the gravity model augmented with regional dummy

variables to examine the determinants of the PRC's trade in parts and components, and to assess the extent of the PRC's trade in parts and components with other East Asian countries.

The paper is organized as follows. Section 2 provides an overview of recent literature to explain the concepts of international production fragmentation. This section also reviews the literature relating to the PRC in the international fragmentation of production. After presenting the scope and sources of the data that we use, Section 3 analyzes the structure of the PRC's parts and components trade. Using the gravity equations, Section 4 presents the determinants of the PRC's trade in parts and components, as compared with the PRC's total trade, manufactured goods trade, and non-manufactured goods trade. Finally, Section 5 summarizes the main findings and concludes the paper.

2. Literature Review

In this section, we review the literature on the concept of international production fragmentation, including literature that looks specifically at the PRC.

2.1. Production Sharing in the World

A lot of research has been conducted on international production sharing since the 1980s. This phenomenon can be described at both the macro and micro levels. From the microeconomic point of view, most intra-industry trade is based on global production sharing on the part of multinational enterprises (MNEs). These firms' production processes have become internationally fragmented as the assembly line has been replaced by different forms of flexible production located in different countries. The various stages of production correspond to different functions and break up the production process into standardized units. This vertical division of production is driven by gains from specialization. From a macro point of view, a country tends to specialize in the individual segments of production in which it has a comparative advantage.

Scholars from different disciplines—economics, management, social economics, economic geography, and other fields—have created different concepts to describe such an economic phenomenon. Jones and Kierzkowski (1990 and 2000) proposed the concept of **production fragmentation**, defined as spatial dispersion of production blocks, which may occur domestically but can also occur across national boundaries. Production fragmentation can take place in vertically integrated companies as well as through outsourcing or market means beyond corporate boundaries. Yi (2003) and Hummels, Ishii, and Yi (2001) put forward the concept of **vertical specialization**, defined as international vertical division of the production process, representing "the imported input content of exports, or equivalently, foreign value added embodied in exports," and provided a calculation method to compute vertical specialization share based exclusively on a country's input–output table. **Intra-product specialization** and **offshore sourcing** were proposed by Arndt (2000). Offshore sourcing arises when a company utilizes the services of another company in a different country to contribute to some portion of their business process in order to reduce costs. **Global production**

sharing was suggested by Yeats (1998) to describe production across national and enterprise borders.²

The above-mentioned studies have focused on four major aspects. The first aspect concerns the extent of international fragmentation of production: which countries are trading in parts and components (and how much) with which other countries? The second concerns the causes of international fragmentation of production: what drives the parts and components trade? The third concerns the economic effects of international fragmentation: specifically, the impact on (i) the overall welfare of trading partners, (ii) factor prices, (iii) the technological level of participating countries, and (iv) income inequality in developed countries. The fourth aspect of the literature is concerned with the role of each country in international production networks and the parts and components trade, and their implications for regional economic integration and cooperation.

The present study is concerned with the first, second, and fourth aspects.

2.2. The People's Republic of China's Role in Production Sharing

As mentioned earlier, there has been no comprehensive study of the PRC's parts and components trade, and most existing studies only focus on a particular trading partner or specific industries.

Amighini (2005) found empirical evidence that during the 1990s the PRC increased its market share in the parts and components trade of information communications technology (ICT) products to rank among the top three ICT exporters in the world. Athukorala and Yamashita (2008) paid attention to global production sharing in their analysis of the United States (US)-PRC trade deficit. They highlighted the two countries' evolving roles in global production networks and separated trade in parts and components, and trade in final goods for the machinery industry. Salleh and Rene (2009) assessed the impact of the PRC's rapid economic growth on machines and telecommunications equipment. Analysis of import trends into the US established that the PRC had affected the export performance of ASEAN5 countries.³ Dean, Loverly, and Mora (2009) examined the pattern of trade between the PRC and its two largest trading partners, Japan and the US, and found that only a small share of these flows could be characterized as arm's length, one-way trade in final goods. Instead, they found extensive two-way trade, deep vertical specialization, concentration of trade in computers and communication devices, and a prominent role for foreign-invested enterprises.

Athukorala (2005, 2009a, 2009b, and 2010) analyzed East Asian export performance and found evidence of the PRC's rise as a global player in the machinery parts and components trade. Kim, Lee, and Park (2010) investigated changes in Asia's regional and global trade linkages, and their influence on relationships among Asia, the EU, and

¹ We will use the above-mentioned terms interchangeably to describe the phenomenon of trade in parts and components.

² ASEAN5 comprises Indonesia, Malaysia, Philippines, Singapore, and Thailand.

the US. They found that the PRC plays a critical role as an assembly and production center in rapidly expanding intra-Asian trade. However, the PRC's share in the parts and components trade involving Europe and the US is also rising, suggesting that East Asia's production chains are becoming increasingly integrated into the global business network.

In summary, the literature finds that the PRC plays an important role in the world's parts and components trade. However, most previous studies are limited to specific industries and the PRC's relationships with specific countries and regions.

In this paper, we look at the role of international processing activities in the PRC's trade—or equivalently, the PRC's role as an assembly and production center—from a macro viewpoint. The PRC's integration into the world economy is one of the most important developments of the global trading system. Specifically, the rapid growth of trade in parts and components is the direct result of the expansion of global production networks. It is thus necessary to give a comprehensive description of the rapid growth of the PRC's trade in parts and components.

3. Structure of the Parts and Components Trade

In this section, we examine the structure of the PRC's parts and components trade. We first look at the overall structure—total parts and components exports, and their share in the PRC's total exports—and then the specific structure—parts and components exports and imports in different product groups. In addition, we assess the relative significance of different countries and regions as export destinations and import sources.

3.1. Trends and Patterns of World Trade in Parts and Components

Data from the UN Comtrade database were utilized, based on Revision 3 of the SITC. SITC Revision 3 marks a significant improvement over SITC Revisions 2 and 1. Apart from providing comprehensive coverage of parts and components in SITC 5 (chemicals and related products), SITC 6 (manufactured goods classified chiefly by material), and SITC 7 (machinery and transport equipment), Revision 3 also separately reports parts and components of some products belonging to SITC 8 (miscellaneous manufactured articles). The classification of products as parts and components of SITC 5, SITC 6, SITC 7, or SITC 8 follows Athukorala (2010).

The earliest year for which trade data are available for all reporting countries is 1992. Therefore, we make use of data from the United Nations (UN) Comtrade database for 1992–2009.

In order to give a complete and comprehensive description of the PRC's international production fragmentation and networks, we selected 144 trade partners of the PRC as a cross-section sample. We also look at a few main regions as regional cross-sections: East Asia, which includes Japan; the Republic of Korea; and Hong Kong, China; and the ASEAN6,⁴ NAFTA, and the EU15.⁵

³ Indonesia, Malaysia, Philippines, Thailand, Singapore, and Viet Nam.



Figure 1: World Trade in Parts and Components (\$ billion)

P&C = parts and components.

Source: Data calculated from UN Comtrade database.

World trade in parts and components and in final goods both show simultaneous growth with total world trade (Figure 1). World trade in parts and components increased more than 3.2 times from about \$961 billion in 1992 to nearly \$3,845 billion in 2008. After the onset of the global economic crisis in 2008, it then decreased to \$3,036 billion in 2009. Parts and components accounted for around 15% of total world trade and did not fluctuate much during 1992–2009. Final goods trade increased 3.1 times from about \$4,258 billion in 1992 to \$16,655 billion in 2008. The share of final goods in total world trade was stable at around 55% throughout the sample period.

3.2. Overall Structure of the People's Republic of China's Trade in Parts and Components

Until the global economic crisis of 2008/09, the PRC's parts and components trade had been increasing steadily in parallel with its total trade since the early 1990s; the PRC's trade of parts and components increased nearly 24 times from about \$16 billion in 1992 to nearly \$430 billion in 2008. In 1992 parts and components accounted for less than 10% of the PRC's total trade, but its share experienced a secular rise until 2006, when it peaked at 21%. However, there was a sharp fall in 2007 and no significant recovery in the next 2 years. The PRC's trade in final goods increased nearly 12 times from about \$115 billion in 1992 to nearly \$1,634 billion in 2008. In contrast to parts and components, the share of final goods slightly declined from 74% in 1994 to 63% in 2006. The share

⁴ Austria, Belgium, Denmark, Finland, France, Germany, Greece, Ireland, Italy, Luxembourg, the Netherlands, Portugal, Spain, Sweden, and the United Kingdom.

recovered modestly in 2007, but fell again after the 2008/09 global economic crisis (Figure 2). The overall trend of convergence between the share of parts and components and the share of final goods supports the notion of the PRC as an assembly and production center.

Comparing the PRC's parts and components trade with the world average, the share of parts and components as a portion of PRC trade flows grew very rapidly for both exports and imports between 1992 and 2009. As a result, the share of parts and components in the PRC's imports was much higher than the world average in 2009. For the PRC's exports, the share of parts and components was far lower than the world average in the early 1990s but it subsequently rose to reach the world average in 2004 (Figure 3).

We now look at the share of parts and components versus final goods in the PRC's exports and imports. Final goods dominated the PRC's exports throughout the sample period. In 1992 final goods accounted for 73.8% of the PRC's exports, rising to reach 80.8% by 2009. The share of parts and components in the PRC's exports also increased, from 4.9% in 1992 to 12.8% in 2009. The pattern for imports was somewhat different from that of exports. The gap between the share of final goods imports and the share of parts and components imports narrowed during 1992–2009. The share of final goods as a portion of total PRC imports increased to 70.9% in 1994, but had fallen to 44.5% by 2009. The share of parts and components steadily increased to 27.9% by 2006, but decreased somewhat in subsequent years (Figure 4).

Figure 2: The People's Republic of China's Trade in Parts and Components (\$ billion)



PRC = People's Republic of China, P&C = parts and components.



Figure 3: Parts and Components Trade as a Share of Total Trade—The People's Republic of China and the World Average

PRC = People's Republic of China, P&C = parts and components. Source: Data calculated from UN Comtrade database.

Figure 4: The People's Republic of China's Exports and Imports—Parts and Components vs. Final Goods



→Share of final goods in export trade → Share of final goods in import trade
→Share of P&C in export trade

PRC = People's Republic of China, P&C = parts and components. Source: Data calculated from UN Comtrade database. Furthermore, we found that the share of final goods in exports (averaging 77.4%) was consistently larger than its share in imports (averaging 55.7%). In contrast, the share of parts and components in the PRC's imports (averaging 21.1%) was larger than its share in exports (averaging 10.5%).

The overall trends suggest that the PRC has become a global production center, importing components and exporting finished products. The rapid growth of the PRC's parts and components exports indicates that the country has also become an increasingly important global supply base for parts and components.

3.3. Structure of the People's Republic of China's Parts and Components Trade

The definition of parts and components includes four main SITCs: SITC 5 (chemicals and related products), SITC 6 (manufactured goods classified chiefly by material), SITC 7 (machinery and transport equipment), and SITC 8 (miscellaneous manufactured articles). The four SITCs each display different characteristics.

As Tables 1a and 1b show, the parts and components that the PRC exports and imports are primarily machinery and transport equipment (SITC 7), chemicals and related products (SITC 5), manufactured goods classified chiefly by material (SITC 6), and miscellaneous manufactures articles (SITC 8).

The share of SITC 7 parts and components in the PRC's total exports increased from 3.8% to 13.9% between 1992 and 2006, but then fell in 2007. Similarly, the share of parts and components in the PRC's SITC 7 exports rose from 24.3% to 32.6% between 1992 and 2002, but then fell in 2003. The share of parts and components in the PRC's trade in the three other aforementioned SITCs is much smaller. The share of parts and components in the PRC's SITC 8 exports increased from 1.7% to 2.7% between 1992 and 2009. For SITC 6 exports, the share of parts and components rose from 1.9% to 3.3%. For SITC 5 exports, the share of parts and components dropped from 0.9% to 0.4%.

In the context of imports, parts and components hold a dominant position for SITC 7. The share of parts and components in the PRC's SITC 8 imports fell sharply from 20.2% to 6.8% between 1992 and 2009. However, the export value of SITC 8 parts and components exceeded the import value, although the export share of parts and components trade was lower than the import share. This reflects the fact that the PRC is a major exporter of SITC 8 finished products. For SITC 5 and SITC 6, the export and import shares of parts and components show similar trends, with both SITCs accounting for relatively small shares of the PRC's parts and components imports.

Year	Total	Expor	t of P&C	Expo	rt of P&C i	n SITC 5	Ехроі	rt of P&C ir	SITC 6	Export	of P&C in	SITC 7	Expor	t of P&C in	SITC 8
	Export Volume of the PRC	Export Value of P&C	Share of P&C in Total Exports (%)	Export Value P&C SITC 5	Share in Total Exports (%)	Share in SITC 5 (%)	Export Value P&C SITC 6	Share in Total Exports (%)	Share in SITC 6 (%)	Export Value P&C SITC 7	Share in Total Exports (%)	Share in SITC 7 (%)	Export Value P&C SITC8	Share in Total Exports (%)	Share in SITC 8 (%)
1992	84.94	4.13	4.9	0.04	0.0	0.9	0.31	0.4	1.9	3.21	3.8	24.3	0.58	0.7	1.7
1993	91.74	4.94	5.4	0.03	0.0	0.6	0.32	0.4	2.0	3.99	4.4	26.1	0.60	0.7	1.6
1994	121.01	6.76	5.6	0.05	0.0	0.9	0.48	0.4	2.1	5.37	4.4	24.5	0.85	0.7	1.7
1995	148.78	9.79	6.6	0.13	0.1	1.4	0.59	0.4	1.8	8.16	5.5	26.0	0.91	0.6	1.7
1996	151.05	10.65	7.1	0.06	0.0	0.6	0.63	0.4	2.2	9.11	6.0	25.8	0.85	0.6	1.5
1997	182.79	13.73	7.5	0.07	0.0	0.7	0.77	0.4	2.2	11.79	6.5	27.0	1.08	0.6	1.6
1998	183.81	16.07	8.7	0.07	0.0	0.7	0.83	0.5	2.6	13.81	7.5	27.5	1.35	0.7	1.9
1999	194.93	19.96	10.2	0.78	0.0	0.7	0.89	0.5	2.7	17.76	9.1	30.2	1.24	0.6	1.7
2000	249.20	27.80	11.2	0.78	0.0	0.6	1.09	0.4	2.6	25.22	10.1	30.5	1.42	0.6	1.7
2001	266.10	32.02	12.0	0.94	0.0	0.6	1.18	0.4	2.7	29.29	11.0	30.9	1.47	0.5	1.7
2002	325.60	44.59	13.7	0.12	0.0	0.6	1.40	0.4	2.6	41.36	12.7	32.6	1.71	0.5	1.7
2003	438.23	61.75	14.1	0.18	0.0	0.6	1.73	0.4	2.5	57.50	13.1	30.6	2.40	0.6	1.9
2004	593.33	87.36	14.7	0.24	0.0	0.7	2.47	0.4	2.5	81.15	13.7	30.3	3.57	0.6	2.3
2005	761.95	112.63	14.8	0.23	0.0	0.7	3.35	0.4	2.6	105.00	13.8	29.8	4.47	0.6	2.3
2006	968.94	145.35	15.0	0.34	0.0	0.8	4.45	0.5	2.5	135.00	13.9	29.6	5.29	0.5	2.2
2007	1220.06	153.59	12.6	0.40	0.0	0.7	5.82	0.5	2.6	140.24	11.5	24.3	7.01	0.6	2.4
2008	1430.69	182.27	12.7	0.34	0.0	0.4	6.75	0.5	2.6	166.05	11.6	24.6	9.37	0.7	2.8
2009	1201.65	153.38	12.8	0.28	0.0	0.5	6.05	0.5	3.3	139.00	11.6	23.5	8.05	0.7	2.7

Table 1a: The People's Republic of China's Export Trade in Parts and Components(\$ billion)

PRC = People's Republic of China, P&C = parts and components, SITC = Standard International Trade Classification.

Year	Total	Import	of P&C	Impor	t of P&C in	SITC5	Impor	t of P&C in	SITC6	Impor	t of P&C in	SITC 7	Impor	t of P&C in	in SITC8				
	Import volume of the PRC	Import value of P&C	Share of P&C in total import (%)	Import value P&C SITC5	Share in total Imports (%)	Share in SITC5 (%)	Import value P&C SITC6	Share in total Imports (%)	Share in SITC6 (%)	Import value P&C SITC7	Share in total Imports (%)	Share in SITC7 (%)	Import value P&C SITC8	Share in total Imports (%)	Share in SITC8 (%)				
1992	84.94	11.90	14.8	0.22	0.3	2.0	0.47	0.6	2.4	10.10	12.5	32.8	1.12	1.4	20.2				
1993	91.74	13.49	13.0	0.34	0.3	3.5	0.50	0.5	1.8	11.51	11.8	25.8	1.14	1.1	18.3				
1994	121.01	15.06	13.0	0.39	0.3	3.3	0.59	0.5	2.1	12.93	11.2	25.1	1.14	1.0	17.0				
1995	148.78	18.76	14.2	0.43	0.3	2.5	0.70	0.5	2.4	16.36	12.4	31.1	1.26	0.9	15.5				
1996	151.05	21.34	15.4	0.45	0.3	2.5	0.72	0.5	2.3	18.88	13.6	34.5	1.30	0.9	15.5				
1997	182.79	25.39	17.8	0.43	0.3	2.2	0.77	0.5	2.4	22.79	16.0	43.2	1.39	1.0	16.3				
1998	183.81	29.00	20.7	0.40	0.3	2.0	0.81	0.6	2.6	26.33	18.8	46.3	1.45	1.0	17.2				
1999	194.93	36.62	22.1	0.42	0.3	1.7	0.93	0.6	2.7	33.82	20.4	48.7	1.44	0.9	14.9				
2000	249.20	52.00	23.1	0.41	0.2	1.4	1.16	0.5	2.8	48.75	21.7	53.0	1.68	0.8	13.2				
2001	266.10	56.92	23.4	0.42	0.2	1.3	1.30	0.5	3.1	53.56	22.0	50.0	1.64	0.7	10.9				
2002	325.60	75.43	25.6	0.45	0.2	1.2	1.77	0.6	3.7	71.30	24.2	52.0	1.91	0.7	9.7				
2003	438.23	109.06	26.4	0.48	0.1	1.0	2.56	0.6	4.0	103.00	24.9	53.4	2.65	0.6	8.0				
2004	593.33	148.02	26.4	0.54	0.1	0.8	3.40	0.6	4.6	141.00	25.1	55.8	3.36	0.6	6.7				
2005	761.95	178.38	27.0	0.59	0.1	0.8	3.87	0.6	4.8	170.00	25.8	58.5	3.60	0.5	5.9				
2006	968.94	220.56	27.9	0.64	0.1	0.7	4.77	0.6	5.5	211.00	26.7	59.1	4.12	0.5	5.8				
2007	1220.06	238.50	24.9	0.66	0.1	0.6	5.66	0.6	5.5	227.31	23.8	55.1	4.88	0.5	5.6				
2008	1430.69	248.23	21.9	0.66	0.1	0.6	6.08	0.5	5.7	235.46	20.8	53.3	6.03	0.5	6.2				
2009	1201.65	227.80	22.7	0.57	0.1	0.5	5.46	0.5	5.1	216.02	21.5	52.9	5.75	0.6	6.8				

Table 1b: The People's Republic of China's Import Trade in Parts and Components (\$ billion)

PRC = People's Republic of China, P&C = parts and components, SITC = Standard International Trade Classification.

Tables 2a and 2b make use of the United Nations (UN) 2-digit product codes within SITC 5, SITC 6, SITC 7, and SITC 8 in order to assess the relative importance of individual product groups in the PRC's trade. The tables describe each item and provide their respective value and share in a particular year. This allows us to see the trends and patterns at a deeper level of fragmentation. We chose four observation years: 1992, 1999, 2006, and 2009.

Machinery and transport equipment (SITC 7) accounted for 77.6% of the PRC's parts and components exports in 1992 and 93.0% in 2006, although the share fell modestly to 90.6% in 2009. For imports, the share increased from 84.8% in 1992 to 95.7% in 2006, and then decreased to 94.8% in 2009. SITC 7 accounts for a high proportion of both exports and imports throughout the sample period. This suggests that trade in SITC 7 parts and components plays a major role in the PRC's participation in international production networks.

Within SITC 7, the most important 2-digit product codes are office machinery (SITC 75), telecommunication and sound equipment (SITC 76), and electrical machinery and apparatus (SITC 77). In 2006, SITC 75, 76, and 77 accounted for 23.9%, 21.7%, and 31.9% of the PRC's parts and components imports, respectively, and 8.9%, 11.4%, and 65.1% of parts and components exports, respectively.

Taking a closer look, electrical machinery and apparatus (SITC 77) accounted for the largest share of the PRC's parts and components trade. SITC 77's share in imports and exports of parts and components reached 85.4% and 40.4%, respectively, in 2009.

The parts and components trade in SITC 6 (62, 65, 66, and 69) and SITC 8 (81, 82, 87, and 88) increased significantly between 1992 and 2009, although their shares of the PRC's total parts and components trade remained relatively small.

SITC code	Product	1992		1999	1999		6	2009	
Exports		\$ million	%	\$ million	%	\$ million	%	\$ million	%
5	Chemicals and related products	37.1	0.9	68.2	0.3	227.0	0.2	279.9	0.2
6	Manufactured goods classified chiefly by material	307.4	7.4	892.3	4.5	4,452.2	3.1	6,047.7	3.9
62	Rubber manufactures	76.4	1.8	236.8	1.2	1,167.4	0.8	1,517.2	1.0
65	Textile yarn, fabric	65.6	1.6	122.6	0.6	492	0.3	1,017.6	0.7
66	Non-metal, mineral manufactures	28.1	0.7	168.8	0.8	975.7	0.7	1,357.0	0.9
69	Metals manufactures	137.3	3.3	364.2	1.8	1,817.1	1.3	2,155.9	1.4
7	Machinery and transport equipment	3,208	77.6	17,756.1	89.0	135,400.2	93.0	139,003.1	90.6
71	Power generating-machines	213.1	5.2	473.2	2.4	2,984.1	2.1	8,395.0	5.5
72	Special industrial machinery	143.7	3.5	419.5	2.1	3,341.5	2.3	5,022.8	3.3
73	Metalworking machinery	30.6	0.7	124.2	0.6	934.0	0.6	1,679.3	1.1
74	General industrial machinery nets	153.5	3.7	813.9	4.1	7,370.5	5.1	11,732.7	7.6
75	Office machines	565.7	13.7	4,126.2	20.7	34,611.2	23.9	26,122.3	17
76	Telecommunication and sound equipment	706.8	17.1	3,735.0	18.7	31,473.9	21.7	13,371.8	8.7
77	Electrical machinery and apparatus	1,132.7	27.4	7,029.2	35.2	46,236.1	31.9	61,899.7	40.4
78	Road vehicles	231	5.6	1,327.1	6.7	11,507.6	7.9	14,771.7	9.6
79	Other transport equipment	99.6	2.4	351	1.8	1291	0.9	1,243.2	0.8
8	Miscellaneous manufactured articles	581.7	14.1	1,238.5	6.2	5,294.6	3.6	8,049.5	5.2
81	Prefabricated buildings fixtures and fittings	59.9	1.4	133.1	0.7	647.8	0.4	1,342.5	0.9
82	Furniture and parts thereof	7.2	0.2	39.8	0.2	860.9	0.6	1,600.9	1.0
84	Apparel and clothing accessories	42.7	1.0	129.3	0.6	229.4	0.2	280.6	0.2
87	Professional, scientific instruments.	25.4	0.6	193	1.0	1,038.9	0.7	1,600.1	1.0
88	Photographic apparatus, optical goods, and watches and clocks	333.3	8.1	511.9	2.6	1,868.9	1.3	1,997.5	1.3
89	Miscellaneous manufactured articles	113.1	2.7	231.4	1.2	648.7	0.4	1,228.0	0.8
P&C	Total	4,134.2	100	19,955.1	100	145,374.0	100	153,380.1	100

Table 2a: Percentage of Parts and Components in the People's Republic of China's Exportsby 2-Digit SITC Categories

Note: Tables 2a uses 5-digit Standard International Trade Classification (SITC) parts and components (P&C) items aggregated at the 2-digit level.

SITC code Product		1992	2	1999	9	2006	;	2009	
Exports		\$ million	%	\$ million	%	\$ million	%	\$ million	%
5	Chemicals and related products	222.0	1.9	419.4	1.1	644.0	0.3	572.3	0.3
6	Manufactured goods classified chiefly by material	465.9	3.9	932.7	2.5	4,774.0	2.2	5,460.1	2.4
62	Rubber manufactures	49.0	0.4	214.7	0.6	1,284.6	0.6	1,649.4	0.7
65	Textile yarn, fabric	199.4	1.7	206.0	0.6	489.3	0.2	530.6	0.2
66	Non-metal, mineral manufactures	22.2	0.2	114.4	0.3	894.1	0.4	917.6	0.4
69	Metals manufactures	195.3	1.6	397.7	1.1	2,106.0	1.0	2,362.6	1.0
7	Machinery and transport equipment	10,096.9	84.8	33,820.8	92.4	211,016.0	95.7	216,017.3	94.8
71	Power generating-machines	1,107.1	9.9	2,495.2	7.1	5,263.2	2.4	5,869.0	2.6
72	Special industrial machinery	966.1	8.6	1,242.3	3.5	3,204.8	1.5	4,299.7	1.9
73	Metalworking machinery	265.5	2.4	376.7	1.1	1,825.6	0.8	1,830.2	0.8
74	General industrial machinery	591.7	5.3	1,179.7	3.3	4,786.0	2.2	6,724.1	3.0
75	Office machines	682.1	5.7	4,196.6	11.5	19,631.5	8.9	13,580.4	6.0
76	Telecommunication and sound equipment	1,843.2	15.5	5,363.7	14.6	25,088.4	11.4	8,769.8	3.8
77	Electrical machinery and apparatus	3,205.8	26.9	17,801.1	48.6	143,476.7	65.1	162,804.7	71.5
78	Road vehicles	1,047.2	8.8	1,470.5	4.0	9362.0	4.2	12,696.8	5.6
79	Other transport equipment	633.9	5.3	871.4	2.4	1,903.6	0.9	2,353.1	1.0
8	Miscellaneous manufactured articles	1,117.7	9.4	1,443.3	3.9	4,124.3	1.9	5,748.5	2.5
81	Prefabricated buildings fixtures and fittings	49.0	0.4	28.7	0.1	53.4	0.0	376.8	0.2
82	Furniture, and parts thereof	4.8	0.0	20.5	0.1	377.6	0.2	609.6	0.3
84	Apparel and clothing accessories	58.3	0.5	115.7	0.3	62.3	0.0	36.6	0.0
87	Professional, scientific instruments.	192.3	1.6	292.5	0.8	1,796.6	0.8	2,670.1	1.2
88	Photographic apparatus, optical goods and watches and clocks	662.5	5.6	911.1	2.5	1,663.7	0.8	1,813.3	0.8
89	Miscellaneous manufactured articles,	150.8	1.3	74.8	0.2	170.8	0.1	242.1	0.1
P&C	Total	11,902.5	100	36,616.3	100	220,558.0	100.0	227,798.2	100

Table 2b: Percentage of Parts and Components in the People's Republic of China's Importsby 2-Digit Categories

Note: Table 2b uses 5-digit Standard International Trade Classification (SITC) parts and components (P&C) items aggregated at the 2-digit level.

3.4. The People's Republic of China's Parts and Components Trade— Destination and Source Countries

The export destination countries and import source countries of the PRC's parts and components trade are heavily concentrated (Figures 5 and 6; Tables 3a and 3b). The PRC's most important trading partners for parts and components are Japan; the Republic of Korea; Hong Kong, China; ASEAN; the EU15; and NAFTA. NAFTA and Hong Kong, China are the main destinations for the PRC's exports, while Japan and ASEAN are the main sources of imports. Meanwhile, the shares of imports accounted for by Hong Kong, China and Japan fell sharply between 1992 and 2009. In contrast, the shares of ASEAN countries and the Republic of Korea rose significantly over the same period. Hong Kong, China, which is the most important destination for the PRC's parts and components exports, is not a significant source of the PRC's imports.

The PRC's parts and components trade is by no means limited to its major trading partners. The market share of the rest of the world increased during 1992–2009 for both imports and exports to reach substantial levels. This trend suggests that the PRC's parts and components trade is an integral part of increasingly globalized production networks.



Figure 5: Export Destination Countries for the Parts and Components Trade of the People's Republic of China

ASEAN = Association of Southeast Asian Nations, EU = European Union, NAFTA = North American Free Trade Area, PRC = People's Republic of China.



Figure 6: Import Source Countries for the Parts and Components Trade of the People's Republic of China

ASEAN = Association of Southeast Asian Nations, EU = European Union, NAFTA = North American Free Trade Area, PRC = People's Republic of China.

Source: Data calculated from UN Comtrade database.

Table 3a: The People's Republic of China's Parts and Components ExportShare by Major Trading Partners and Regions

Reporting Country or Region	1992	1999			2006		2009		
	\$ million	%	\$ million	%	\$ million	%	\$ million	%	
Japan	292.5	7.1	3221.7	16.1	13,779.8	9.5	12,226.4	7.9	
Republic of Korea	53.5	1.3	978.5	4.9	7,730.6	5.3	8,038.1	5.3	
Hong Kong, China	2,465.1	59.6	4,887.5	24.5	39,227.0	27.0	36,735.0	24.0	
ASEAN6	192.3	5.5	2,428.3	12.2	16,536.2	11.4	17,132.2	11.2	
NAFTA	314.0	7.6	3,527.8	17.7	25,709.9	17.7	23,064.8	15.0	
EU15	197.7	4.8	2,384.1	11.9	19,395.3	13.3	23,012.6	15.0	
Total	3,515.0	85.8	17,427.9	87.3	122,378.7	84.2	120,208.9	78.4	
World	4,134.2	100	19,955.1	100	145,374.0	100	153,380.1	100	

ASEAN = Association of Southeast Asian Nations, EU = European Union, NAFTA = North American Free Trade Area. Source: Data calculated from UN Comtrade database.

Reporting Country or Region	1992 \$ million	%	1999 \$ million	%	2006 \$ million	%	2009 \$ million	%
Japan	2,284.1	19.2	10,387.0	28.7	37,879.6	17.2	38,688.7	17.0
Republic of Korea	216.1	1.8	3,370.4	9.3	33,511.8	15.2	33,472.4	17.0
Hong Kong, China	4,576.8	38.5	1,947.4	5.4	3,155.1	1.4	2,260.8	1.0
ASEAN6	97.6	0.8	3,782.7	10.5	38,993.7	17.7	34,758.5	15.3
NAFTA	1,204.6	10.1	4,456.2	12.3	15,656.7	7.1	14,992.9	6.6
EU15	2,031.3	17.1	6,150.5	17.0	18,483.1	8.4	23,206.4	10.2
Total	10,410.5	87.5	30,094.2	83.2	147,680.0	67.0	147,379.6	64.7
World	11,902.6	100	36,161.3	100	220,558.0	100	227,798.0	100

Table 3b: The People's Republic of China's Parts and Components ImportShare by Major Trading Partners and Regions

ASEAN = Association of Southeast Asian Nations, EU = European Union, NAFTA = North American Free Trade Area. Source: Data calculated from UN Comtrade database.

4. The Determinants of the People's Republic of China's Trade in Parts and Components

In this section, we estimate the gravity equation—a widely used empirical tool for analyzing international trade flows—to identify the main determinants of the PRC's parts and components exports and imports. In addition to parts and components, we also estimate the gravity equation for the other major groups of traded goods—manufactured goods, non-manufactured goods, and final goods—for comparative purposes.

4.1. The Gravity Equation

Tinbergen (1962) and Pöyhönen (1963) pioneered the simple gravity equation in which the volume of trade between two countries is proportional to the product of their masses (i.e., gross domestic product [GDP]) and inversely related to the distance between them. The gravity equation has enjoyed widespread empirical success. Recently, renewed interest in geography among economists has rekindled its popularity as a tool for empirical analysis. Furthermore, a number of studies have shown that the gravity equation can be derived from many different models of international trade (Helpman and Krugman 1985, Bergstrand 1989, Deardorff 1998, Eaton and Kortum 2002, and Evenett and Keller 2002).⁶ Therefore, it possesses "more theoretical foundation than any other trade model" (Baldwin 2006).

⁵ Harrigan (2001) and Anderson and van Wincoop (2003) provide a comprehensive review of the literature on the theoretical foundations for the gravity model. Baier et al. (2007) address the potential problems in estimating the gravity model to isolate the effects of a free trade agreement on bilateral trade.

The standard gravity equation takes the following form:

 $LnXMijt = \alpha + \beta 1LnGDPit + \beta 2LnGDPjt + \beta 3LnDISTij + \epsilon ijt,$

where

LnXM*ijt* = log of export (or import) flows from country *i* to country *j* at time t LnGDP*it* = log of GDP of country *i* at time t LnGDP*jt* = log of GDP of country *j* at time t LnDIST*ij* = log of geographical distance between country *i* and country *j* ϵijt = random disturbance term

4.2. The Augmented Gravity Equation for the People's Republic of China's Trade in Parts and Components

Based on the equation above, dummy variables are introduced for countries sharing a common border, countries surrounded by land, and language commonality. We also include a dummy variable for World Trade Organization (WTO) member countries and another dummy variable to take into account the PRC's bilateral and sub-regional free trade agreements (FTAs):

LnXM*ijt* =
$$\alpha$$
+ β 1LnGDP*it* + β 2LnGDP*jt* + β 3LnDIST*j*+ β 4CONTIG*ij* + β 5LANDLOCKED*ij* + β 6COMLANG*ij* + β 7WTO*jt* + β 8FTA*ijt* + *ui* + *ut* + ϵ *ijt*,

where

LnXM: bilateral trade (total manufactured goods or parts and components under SITC5, SITC6, SITC 7, and SITC 8)

LnGDPit = log of GDP of country i (PRC) at time t

 $LnGDP_{jt} = log of GDP of country j at time t$

- LnDISTij = log of geographical distance between country *i* (PRC) and country *j*
- CONTIG ij=1 if economy *i* (PRC) and economy *j* share the same border = 0 otherwise
- LANDLOCKED*j*=1 if economy *j* is a landlocked economy
 - = 0 otherwise
- COMLANGij = 1 if economy *i* (PRC) and economy *j* share the same language = 0 otherwise
- WTOjt = 1 if economy *j* is a WTO member at time t
 - = 0 otherwise
- FTAijt = 1 if economy *i* (PRC) and economy *j* are members of a bilateral or sub-regional FTA at time t
 - = 0 otherwise

Various regional dummies may also be included.

We estimate the equation separately for imports and exports. For both imports and exports, we again estimate the equation separately for all goods, manufactured goods, non-manufactured goods, final goods, and parts and components.

Our data cover the PRC's trade with 144 countries during 1992–2007. The augmented gravity model is first estimated using the most common fixed effects procedure to account for time-invariant partner country fixed effects. We also include year dummies to take account of factors such as world business cycles and global financial shocks. In the fixed effects model, we cannot estimate time-invariant variables such as distance and regional dummies. Therefore, as an alternative, we also estimate a random effects model with the assumption of strict exogeneity and orthogonality between explanatory variables and the error term.

4.3 Estimation Results

Tables 4 and 5 report the results obtained from the fixed effects model for the PRC's exports and imports of various types of goods, respectively.

There are four significant points to note about the results in Table 4. First, the estimated coefficients for the PRC's GDP are positive and significant; a 10% increase in the PRC's GDP is associated with a 15% increase in total exports (Column I). When total exports are divided into manufactured products and non-manufactured products, manufactured exports are more closely associated with the PRC's GDP. Specifically, a 10% increase in the PRC's GDP is associated with a 16% increase in the export of manufactured products (Column II) and an 8% increase in the export of non-manufactured products (Column III). This contradicts the widespread expectation that the PRC will increasingly export more manufactured products as it grows. When exports of manufactured products are divided into final products and parts and components, we observe that their response to the PRC's GDP growth is about the same for both types of manufactured products (Columns IV and V).

Second, the estimated coefficients for partner countries' GDP are also positive and significant, in line with theoretical predictions. More precisely, a 10% increase in partner countries' GDP is associated with a 10% increase in the PRC's total exports. Furthermore, the foreign income elasticity of PRC exports is greater for manufactured products than for non-manufactured products, and greater for parts and components than for final products.

Third, the results indicate that the PRC exports more to WTO members than to nonmembers. Specifically, the PRC exports 39% more to WTO members than to nonmembers.⁷ The WTO membership effect is greater for non-manufactured products than for manufactured products, and greater for parts and components than for final products.

Fourth, the estimated coefficients for an FTA are negative and significant, which implies that the PRC does not export more to its FTA partners.

⁶ Calculated as $39\% = (\exp(.329)-1)100$.

	I.	II	III	IV	v
	Total	Manufactured	Non-Manufactured	Final	Parts and Components
Lgdp_h	1.455***	1.628***	0.811***	1.631***	1.823***
	(0.044)	(0.046)	(0.088)	(0.046)	(0.078)
Lgdp_p	1.040***	1.045***	0.723***	1.021***	1.287***
	(0.058)	(0.060)	(0.114)	(0.061)	(0.102)
WTO	0.329***	0.274***	0.589***	0.263***	0.588***
	(0.060)	(0.062)	(0.121)	(0.063)	(0.111)
FTA	-0.252**	-0.230**	-0.335*	-0.294***	-0.094
	(0.103)	(0.107)	(0.201)	(0.108)	(0.178)
_cons	-60.305***	-65.388***	-37.842***	-51.180***	-65.860***
	(1.329)	(1.381)	(2.632)	(1.398)	(2.334)
Number of observations	2,242	2,242	2,170	2,242	2,185
R ²	0.780	0.787	0.315	0.777	0.669

Table 4: Determinants of the People's Republic of China's Exports—Fixed Effects Model

Notes: Estimates are made with partner country fixed effects and year dummies included. Standard errors are in parentheses. ***, **, and * show 1%, 5%, and 10% significance, respectively

Source: Author's observations.

	I	Ш	111	IV	v
	Total	Manufactured	Non-Manufactured	Final	Parts and Components
Lgdp_h	1.682***	1.296***	1.949***	1.067***	1.521***
	(0.129)	(0.144)	(0.131)	(0.147)	(0.194)
Lgdp_p	0.372**	0.297	0.239	0.373*	0.502*
	(0.169)	(0.189)	(0.171)	(0.193)	(0.258)
WTO	-0.704***	-0.251	-0.240	-0.295	-0.603**
	(0.177)	(0.192)	(0.184)	(0.199)	(0.273)
FTA	0.025	0.147	-1.023***	0.034	0.723**
	(0.297)	(0.319)	(0.292)	(0.318)	(0.358)
_cons	-51.420***	-40.687***	-56.826***	-22.637***	-40.304***
	(3.954)	(4.383)	(4.013)	(4.434)	(5.902)
Number of observations	2,120	1,996	1,965	1,915	1,459
R ²	0.335	0.205	0.379	0.176	0.190

Table 5: Determinants of the People's Republic of China's Imports—Fixed Effects Model

Notes: Estimates are made with partner country fixed effects and year dummies included. Standard errors are in parentheses. ***, **, and * show 1%, 5%, and 10% significance, respectively.

Source: Author's observations.

The results for PRC imports in Table 5 reveal four major highlights. First, the estimated coefficients for the PRC's GDP are again positive and significant; a 10% increase in the PRC's GDP is associated with a 17% increase in total imports (Column I). Thus, the PRC's GDP growth has a disproportionately positive impact on its demand for foreign goods.

When total imports are divided into manufactured products and non-manufactured products, non-manufactured imports are more closely associated with the PRC's GDP. Specifically, a 10% increase in the PRC's GDP is associated with a 19% increase in imports of non-manufactured products (Column III), and a 13% increase in imports of manufactured products (Column II). When imports of final products and parts and components are estimated separately, we observe that both types of imports increase by about the same percentage in response to increases in the PRC's GDP (Columns IV and V).

Second, the estimated coefficient for partner countries' GDP is also positive and significant, but the size of the coefficient is considerably smaller than the coefficient for the PRC's GDP. For example, a 10% increase in partner countries' GDP is associated with only a 4% increase in total imports. This is consistent with economic intuition. Furthermore, when total imports are divided, foreign income elasticity is statistically significant only for parts and components.

Third, in direct contrast to the results for PRC exports, Table 5 reveals that the PRC has been importing less from WTO members than from non-members. This is inconsistent with economic intuition and merits further investigation in the future.

Fourth, the PRC does not seem to import more from its FTA partners. When total imports are divided into manufactured and non-manufactured products, the PRC in fact imports less non-manufactured products from its FTA partners. Yet when imports of final products and parts and components are estimated separately, the estimated coefficients for FTA are positive and significant for both.

We now assess whether the PRC's trade in parts and components is greater with countries in East Asia than in other regions. In order to do this, we have included four different Asia dummies: (i) ASEAN6 (Brunei Darussalam, Indonesia, Malaysia, Philippines, Singapore, and Thailand); (ii) Hong Kong, China; (iii) Japan; and (iv) the Republic of Korea. We also include two extra-regional dummies: (i) EU15 (the EU's 15 original member countries) and (ii) NAFTA (US, Canada, and Mexico).

In this case, the fixed effects procedure is not appropriate, since these time-invariant regional dummies cannot be estimated. Therefore, the random effects procedure is employed. In the random effects model, other common gravity variables such as geographic distance are also included.

The results from the random effects procedure are reported in Tables 6 and 7 for PRC exports and imports, respectively.

Focusing first on PRC exports in Table 6, the gravity model works well for all equations, as indicated by the large size of R^2 . The PRC exports more to larger countries and countries that share common borders and a common language, and exports less to landlocked countries. On the other hand, geographic distance does not seem to matter with respect to the PRC's exports. This is in contrast with theoretical predictions and merits further investigation.

In line with the results from fixed effects estimation (Table 4), the PRC exports more to WTO member economies but less to its FTA partners.

Let us now assess whether the PRC exports more to the East Asian region, even after all possible natural and institutional causes of trade are controlled for. As shown in Column I, the PRC exports more to ASEAN6 countries; Hong Kong, China; and the Republic of Korea; but not to Japan. It also exports more to NAFTA member countries and less to EU15 member countries. When total exports are split into manufactured and non-manufactured products, the PRC exports more manufactured products to ASEAN6 countries and Hong Kong, China and exports more non-manufactured products to ASEAN6 countries, Japan, and the Republic of Korea. Interestingly, when exports of final products and parts and components are estimated separately, the PRC exports more final products to ASEAN6 countries; Hong Kong, China; and the Republic of Korea. Thus, the PRC's trade linkages with East Asian countries, in terms of exports, are stronger for the parts and components trade than for final goods.

Table 7 presents the estimated results from the random effects procedure for the PRC's imports.

Again, the gravity model works well for all equations, as indicated by the large size of R². The PRC imports more from larger countries and countries that share common borders and a common language, and imports less from landlocked countries. In contrast to exports, geographic distance is also significant and negative. Similar to the fixed effects results (Table 5), the PRC does not import more from WTO member economies or from its FTA partners.

The PRC imports more from ASEAN6 countries, but not from Hong Kong, China; Japan; or the Republic of Korea. Furthermore, the PRC does not import more from NAFTA member countries and it imports less from EU15 countries.

When total imports are split into manufactured and non-manufactured products, the PRC imports more manufactured products from ASEAN6 countries and less from Hong Kong, China. In addition, the PRC does not import more manufactured products from Japan or the Republic of Korea. On the other hand, the PRC imports more from NAFTA countries and the EU15.

Finally, when imports of final products and parts and components are estimated separately, the PRC does not import more final products from Hong Kong, China while it does import more parts and components from ASEAN6 countries, Japan, and the Republic of Korea. Interestingly, among these four regions of East Asia, the Republic of Korea yields the largest coefficient.

Thus, summarizing the results from Tables 6 and 7, we can conclude that the PRC's trade linkages with East Asian countries in terms of both exports and imports are stronger for parts and components than for final goods.

	I	Ш	ш	IV	V
	Total	Manufactured	Non- Manufactured	Final	Parts and Components
Lgdp_h	1.498***	1.669***	0.783***	1.667***	1.935***
	(0.050)	(0.050)	(0.096)	(0.051)	(0.083)
Lgdp_p	0.912***	0.925***	0.803***	0.915***	0.955***
	(0.040)	(0.042)	(0.049)	(0.042)	(0.049)
Ldistw	-0.169	-0.122	-0.976***	-0.126	-0.276
	(0.225)	(0.228)	(0.230)	(0.225)	(0.276)
Contig	1.342***	1.256***	1.878***	1.292***	1.053**
	(0.441)	(0.475)	(0.344)	(0.469)	(0.517)
Landlocked	-0.604**	-0.574**	-1.421***	-0.604**	-0.563*
	(0.280)	(0.286)	(0.331)	(0.286)	(0.328)
comlang_off	1.064***	1.081***	1.011***	1.059***	1.579***
	(0.342)	(0.331)	(0.322)	(0.318)	(0.469)
WTO	0.330***	0.279***	0.483***	0.269***	0.589***
	(0.081)	(0.077)	(0.147)	(0.076)	(0.152)
FTA	-0.263***	-0.240***	-0.338***	-0.303***	-0.125
	(0.052)	(0.057)	(0.079)	(0.053)	(0.089)
EU15	-0.401**	-0.358*	-0.193	-0.334*	-0.301
	(0.200)	(0.195)	(0.205)	(0.184)	(0.291)
NAFTA	0.307**	0.189	0.428***	0.151	0.672***
	(0.149)	(0.150)	(0.153)	(0.142)	(0.243)
ASEAN6	0.881***	0.815***	1.370***	0.719***	1.501***
	(0.283)	(0.275)	(0.274)	(0.267)	(0.425)
Japan	0.493	0.386	1.147**	0.401	0.641
	(0.617)	(0.604)	(0.538)	(0.616)	(0.655)
Hong Kong, China	1.473*	1.663*	-0.014	1.595	1.913**
	(0.854)	(0.928)	(0.611)	(0.972)	(0.847)
Korea, Rep. of	1.073*	0.962	1.231**	0.936	1.304*
	(0.626)	(0.642)	(0.567)	(0.635)	(0.791)
_cons	-56.973***	-62.630***	-30.172***	-48.587***	-58.699***
	(2.662)	(2.695)	(3.362)	(2.659)	(3.711)
Number of observations	2,242	2,242	2,170	2,242	2,185
R ²	0.836	0.836	0.720	0.835	0.707

Table 6: Determinants of the People's Republic of China'sExports—Random Effects Model

ASEAN = Association of Southeast Asian Nations, EU = European Union, NAFTA = North American Free Trade Area, PRC = People's Republic of China, WTO = World Trade Organization.

Notes: Estimates are made with random effects model. Robust standard errors are in parentheses. ***, **, and * show 1%, 5%, and 10% significance, respectively.

Source: Author's observations.

	I	Ш	Ш	IV	V
	Total	Manufactured	Non- Manufactured	Final	Parts and Components
Lgdp_h	1.328***	0.821***	1.689***	0.633***	1.084***
	(0.129)	(0.132)	(0.122)	(0.134)	(0.191)
Lgdp_p	1.232***	1.493***	0.913***	1.473***	1.293***
	(0.077)	(0.081)	(0.078)	(0.084)	(0.108)
Ldistw	-0.953**	-1.722***	0.014	-2.001***	-1.373*
	(0.422)	(0.382)	(0.420)	(0.434)	(0.710)
Contig	1.362***	1.642***	2.565***	1.489***	-0.021
	(0.478)	(0.499)	(0.572)	(0.554)	(0.953)
Landlocked	-0.224	-0.503	-0.566	-0.700	-0.688
	(0.467)	(0.447)	(0.576)	(0.491)	(0.653)
comlang_off	1.814***	2.205***	1.712***	2.310***	2.805**
	(0.430)	(0.566)	(0.538)	(0.636)	(1.279)
WTO	-0.483***	-0.204	-0.214	-0.309	-0.301
	(0.165)	(0.182)	(0.216)	(0.199)	(0.266)
FTA	0.151	0.289	-0.874***	0.151	0.836**
	(0.235)	(0.241)	(0.148)	(0.224)	(0.397)
EU15	0.164	1.093***	-0.439	1.147***	3.244***
	(0.282)	(0.304)	(0.355)	(0.317)	(0.513)
NAFTA	-0.123	0.737***	-0.382	0.610***	2.285***
	(0.227)	(0.238)	(0.295)	(0.229)	(0.625)
ASEAN6	1.237***	1.043**	2.641***	0.571	3.323***
	(0.394)	(0.523)	(0.420)	(0.670)	(1.189)
Japan	-0.272	-0.960	0.491	-1.378	2.751**
	(0.814)	(0.827)	(0.889)	(0.920)	(1.330)
Hong Kong, China	-1.081	-1.804*	-1.621	-2.279*	2.297
	(1.132)	(0.969)	(1.179)	(1.172)	(2.030)
Korea, Rep. of	1.079	0.476	2.564**	-0.056	3.633**
	(1.008)	(0.999)	(1.160)	(1.103)	(1.755)
_cons	-54.053***	-41.245***	-66.405***	-19.476***	-36.650***
	(5.354)	(5.373)	(5.091)	(5.596)	(8.674)
Number of observations	2,120	1,996	1,965	1,915	1,459
R ²	0.707	0.762	0.549	0.757	0.692

Table 7: Determinants of the People's Republic of China's Imports—Random Effects Model

ASEAN = Association of Southeast Asian Nations, EU = European Union, NAFTA = North American Free Trade Area, PRC = People's Republic of China, WTO = World Trade Organization.

Notes: Estimates are made with random effects model. Robust standard errors are in parentheses. ***, **, and * show 1%, 5%, and 10% significance, respectively.

Source: Author's observations.

5. Concluding Observations

The rapid integration of the PRC into global production sharing has become an important feature of global trade, development, and production networks. The purpose of this paper is to examine the characteristics of the PRC's trade in terms of production sharing, and to provide a more complete description of the PRC's trade in parts and components. We observed and analyzed the PRC's parts and components trade, and then we used an augmented gravity model to identify its main determinants.

At a broader level, given the central importance of the parts and components trade in the PRC's overall trade, and given the central importance of trade in the PRC's growth and development, we believe that our study can substantially deepen and broaden our understanding of the PRC's trade, growth, and development. More specifically, our study helps us to better understand the primary drivers of the PRC's parts and components trade, as well as the structure of this trade by identifying, for example, the most important product groups and trading partners. While the literature contains a large number of studies that look at the issue of the PRC's parts and components trade from the perspective of specific industries or partner countries, what is lacking is a more comprehensive study that provides an overall picture. We hope that our study marks a useful step in filling this major gap in the literature.

In terms of empirical results, we found that the shares of parts and components in the PRC's exports and imports have grown very rapidly. Their share in the PRC's imports is much higher than the world average. In contrast, their share in the PRC's exports was far below the world level in the early 1990s but has since increased to reach the world average. Thus, the parts and components trade has become an increasingly significant element of the PRC's trade.

Significantly, the PRC's imports of parts and components far exceed its exports, confirming the PRC's role as a major world production center. At the same time, the rapid growth of the PRC's parts and components exports shows that the PRC is also becoming an increasingly important manufacturing base for parts and components. The data indicate that SITC 77 (electrical machinery and apparatus) is a pivotal sector in the PRC's parts and components trade, accounting for a large share of both exports and imports.

We also find that the PRC's parts and components trade has a relatively high degree of regional concentration. Japan; Hong Kong, China; the Republic of Korea; ASEAN; NAFTA; and the EU are the PRC's major trading partners. Exports and imports show different structural characteristics. Notwithstanding the high regional concentration, parts and components trade with the rest of the world is also large and growing in relative importance.

Another finding is that the degree of production sharing with the PRC is larger for East Asian countries than for NAFTA or the EU. The PRC's trade linkages with East Asian countries, in terms of both exports and imports, are stronger for parts and components than for final goods. The data indicate that the shares of parts and components in the PRC's exports and imports has declined since 2007, even though there was no reduction in the total amount of imports and exports. Significantly, this change in structure happened before the global economic crisis of 2008/09. This begs the following questions: was the decrease caused by the PRC's macroeconomic structural adjustment in 2007 to prevent its economy from overheating? Or does it have any possible relation with the global crisis? These questions should be addressed and studied in greater depth.

The PRC has become a globally significant trading power and investment destination within the context of regional economic integration in East Asia. The growth of the parts and components trade is closely related to the production processes of multinational firms. Therefore, it would also be useful to analyze the driving forces behind the PRC's parts and components trade at the firm level.

Finally, the PRC and Taipei, China have close trade relations, and Taipei, China is widely seen as one of the PRC's main economic partners. However, unfortunately the UN Comtrade database does not include data for Taipei, China. Its inclusion in the database would allow for a more complete analysis of the PRC's parts and components trade.

References

- A. Amighini. 2005. [People's Republic of] China in the International Fragmentation of Production: Evidence from the ICT Industry. *European Journal of Comparative Economics.* 2 (2). pp. 203–219.
- J. E. Anderson and E. van Wincoop. 2003. Gravity with Gravitas: A Solution to the Border Puzzle. *American Economic Review*. 93 (1). pp. 170–92.
- S. W. Arndt. 2000. Offshore Sourcing and Production Sharing in Preference Areas. Claremont Colleges Working Paper in Economics. 12.
- P. Athukorala. 2005. Product Fragmentation and Trade Patterns in East Asia. Asian *Economic Papers.* 4 (3). pp. 1–27.
- ———. 2009a. The Rise of [the People's Republic of] China and East Asian Export Performance: Is the Crowding-Out Fear Warranted? *The World Economy*. 32 (2). pp. 234–266.

———. 2009b. Production Networks and Trade Patterns: East Asia in a Global Context. *Departmental Working Papers*. 2009-15. Australian National University, Arndt-Corden Department of Economics.

- ———. 2010. Production Networks and Trade Patterns in East Asia: Regionalization or Globalization? *ADB Working Paper Series on Regional Economic Integration*. 56. Manila: Asian Development Bank.
- P. Athukorala and N. Yamashita. 2006. Production Fragmentation and Trade Integration: East Asia in a Global Context. *North American Journal of Economics and Finance*. 17 (3). pp. 233–256.
- ——. 2008. Global Production Sharing and US–[People's Republic of] China Trade Relations. In L. Song and W. T. Woo, eds. [People's Republic of] China's Dilemma: Economic Growth, the Environment, and Climate Change. ANU E Press.
- S. L. Baier, J. H. Bergstrand, and E. Vidal. 2007. Free Trade Agreements in the Americas: Are the Trade Effects Larger than Anticipated? *The World Economy*. 30 (9). pp. 47–89.
- R. Baldwin. 2006. The Euro's Trade Effects. *ECB Working Paper.* 594. European Central Bank.
- J. H. Bergstrand. 1989. The Gravity Equation in International Trade: Some Microeconomic Foundations and Empirical Evidence. *Review of Economics and Statistics.* 67 (3). pp. 474–81.

- J. M. Dean, M. E. Loverly, and J. Mora. 2009. Decomposing [People's Republic of] China–Japan–US Trade: Vertical Specialization, Ownership, and Organization Form. *Journal of Asian Economics*. 20 (6). pp. 596–610.
- A. Deardorff. 1998. Does Gravity Work in a Neoclassical World? In J. A. Frankel, ed. *The Regionalization of the World Economy*. Chicago: University of Chicago Press. pp. 7–32.
- J. Eaton and S. Kortum. 2002. Technology, Geography, and Trade. *Econometrica.* 70. pp. 1741–80.
- S. J. Evenett and W. Keller. 2002. On Theories Explaining the Success of the Gravity Equation. *Journal of Political Economy*. 110. pp. 281–316.
- J. Harrigan. 2001. Specialization and the Volume of Trade: Do the Data Obey the Laws? In K. Choi and J. Harrigan, eds. *The Handbook of International Trade*. London: Basil Blackwell.
- E. Helpman and P. Krugman. 1985. Market Structure and Foreign Trade. MIT Press.
- D. Hummels, J. Ishii, and K. M. Yi. 2001. The Nature and Growth of Vertical Specialization in World Trade. *Journal of International Economics*. 54 (1). pp.75– 96.
- R. W. Jones and H. Kierzkowski. 1990. The Role of Services in Production and International Trade: A Theoretical Framework. In R. W. Jones and A. O. Krueger, eds. *The Political Economy of International Trade*. Oxford: Basil Blackwell.
- ———. 2000. A Framework for Fragmentation. In S. Arndt and H. Kierzkowski, eds. Fragmentation: New Production Patterns in the World Economy. Oxford: Oxford University Press.
- S. Kim, J-W. Lee, and C-Y. Park. 2010. The Ties That Binds Asia, Europe, and the Unitized States. *ADB Economics Working Paper Series*. 192. Manila: Asian Development Bank.
- P. Pöyhönen. 1963. A Tentative Model for the Volume of Trade between Countries. *Weltwirtschaftliches Archive*. 90. pp. 93–100.
- S. I. Salleh and A. L. Rene. 2009. Export Performances of Selected Electronics Products: The Case of [People's Republic of] China and ASEAN-5. Asian Social Science. 5 (2). pp.1–8.
- J. Tinbergen. 1962. Shaping the World Economy. New York: Twentieth Century Fund.
- A. Yeats. 1998. Just How Big is Global Production Sharing? *World Bank Policy Research Working Paper.* 1871.

K. M. Yi. 2003. Can Vertical Specialization Explain the Growth of World Trade? *Journal of Political Economics.* 111(1). pp.52–102.

ADB Working Paper Series on Regional Economic Integration*

- 1. "The ASEAN Economic Community and the European Experience" by Michael G. Plummer
- 2. "Economic Integration in East Asia: Trends, Prospects, and a Possible Roadmap" by Pradumna B. Rana
- "Central Asia after Fifteen Years of Transition: Growth, Regional Cooperation, and Policy Choices" by Malcolm Dowling and Ganeshan Wignaraja
- 4. "Global Imbalances and the Asian Economies: Implications for Regional Cooperation" by Barry Eichengreen
- 5. "Toward Win-Win Regionalism in Asia: Issues and Challenges in Forming Efficient Trade Agreements" by Michael G. Plummer
- 6. "Liberalizing Cross-Border Capital Flows: How Effective Are Institutional Arrangements against Crisis in Southeast Asia" by Alfred Steinherr, Alessandro Cisotta, Erik Klär, and Kenan Šehović
- 7. "Managing the Noodle Bowl: The Fragility of East Asian Regionalism" by Richard E. Baldwin
- "Measuring Regional Market Integration in Developing Asia: a Dynamic Factor Error Correction Model (DF-ECM) Approach" by Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Nedelyn Magtibay-Ramos, and Pilipinas F. Quising
- 9. "The Post-Crisis Sequencing of Economic Integration in Asia: Trade as a Complement to a Monetary Future" by Michael G. Plummer and Ganeshan Wignaraja
- 10. "Trade Intensity and Business Cycle Synchronization: The Case of East Asia" by Pradumna B. Rana
- 11. "Inequality and Growth Revisited" by Robert J. Barro
- 12. "Securitization in East Asia" by Paul Lejot, Douglas Arner, and Lotte Schou-Zibell
- 13. "Patterns and Determinants of Cross-border Financial Asset Holdings in East Asia" by Jong-Wha Lee
- 14. "Regionalism as an Engine of Multilateralism: A Case for a Single East Asian FTA" by Masahiro Kawai and Ganeshan Wignaraja

- 15. "The Impact of Capital Inflows on Emerging East Asian Economies: Is Too Much Money Chasing Too Little Good?" by Soyoung Kim and Doo Yong Yang
- "Emerging East Asian Banking Systems Ten Years after the 1997/98 Crisis" by Charles Adams
- 17. "Real and Financial Integration in East Asia" by Soyoung Kim and Jong-Wha Lee
- 18. "Global Financial Turmoil: Impact and Challenges for Asia's Financial Systems" by Jong-Wha Lee and Cyn-Young Park
- 19. "Cambodia's Persistent Dollarization: Causes and Policy Options" by Jayant Menon
- 20. "Welfare Implications of International Financial Integration" by Jong-Wha Lee and Kwanho Shin
- 21. "Is the ASEAN-Korea Free Trade Area (AKFTA) an Optimal Free Trade Area?" by Donghyun Park, Innwon Park, and Gemma Esther B. Estrada
- 22. "India's Bond Market—Developments and Challenges Ahead" by Stephen Wells and Lotte Schou- Zibell
- 23. "Commodity Prices and Monetary Policy in Emerging East Asia" by Hsiao Chink Tang
- 24. "Does Trade Integration Contribute to Peace?" by Jong-Wha Lee and Ju Hyun Pyun
- 25. "Aging in Asia: Trends, Impacts, and Responses" by Jayant Menon and Anna Melendez-Nakamura
- 26. "Re-considering Asian Financial Regionalism in the 1990s" by Shintaro Hamanaka
- 27. "Managing Success in Viet Nam: Macroeconomic Consequences of Large Capital Inflows with Limited Policy Tools" by Jayant Menon
- "The Building Block versus Stumbling Block Debate of Regionalism: From the Perspective of Service Trade Liberalization in Asia" by Shintaro Hamanaka
- 29. "East Asian and European Economic Integration: A Comparative Analysis" by Giovanni Capannelli and Carlo Filippini
- 30. "Promoting Trade and Investment in India's Northeastern Region" by M. Govinda Rao

- 31. "Emerging Asia: Decoupling or Recoupling" by Soyoung Kim, Jong-Wha Lee, and Cyn-Young Park
- 32. "India's Role in South Asia Trade and Investment Integration" by Rajiv Kumar and Manjeeta Singh
- 33. "Developing Indicators for Regional Economic Integration and Cooperation" by Giovanni Capannelli, Jong-Wha Lee, and Peter Petri
- 34. "Beyond the Crisis: Financial Regulatory Reform in Emerging Asia" by Chee Sung Lee and Cyn-Young Park
- 35. "Regional Economic Impacts of Cross-Border Infrastructure: A General Equilibrium Application to Thailand and Lao PDR" by Peter Warr, Jayant Menon, and Arief Anshory Yusuf
- 36. "Exchange Rate Regimes in the Asia-Pacific Region and the Global Financial Crisis" by Warwick J. McKibbin and Waranya Pim Chanthapun
- "Roads for Asian Integration: Measuring ADB's Contribution to the Asian Highway Network" by Srinivasa Madhur, Ganeshan Wignaraja, and Peter Darjes
- 38. "The Financial Crisis and Money Markets in Emerging Asia" by Robert Rigg and Lotte Schou-Zibell
- 39. "Complements or Substitutes? Preferential and Multilateral Trade Liberalization at the Sectoral Level" by Mitsuyo Ando, Antoni Estevadeordal, and Christian Volpe Martincus
- 40. "Regulatory Reforms for Improving the Business Environment in Selected Asian Economies—How Monitoring and Comparative Benchmarking can Provide Incentive for Reform" by Lotte Schou-Zibell and Srinivasa Madhur
- 41. "Global Production Sharing, Trade Patterns, and Determinants of Trade Flows in East Asia" by Prema-chandra Athukorala and Jayant Menon
- 42. "Regionalism Cycle in Asia (-Pacific): A Game Theory Approach to the Rise and Fall of Asian Regional Institutions" by Shintaro Hamanaka
- 43. "A Macroprudential Framework for Monitoring and Examining Financial Soundness" by Lotte Schou-Zibell, Jose Ramon Albert, and Lei Lei Song
- 44. "A Macroprudential Framework for the Early Detection of Banking Problems in Emerging Economies" by Claudio Loser, Miguel Kiguel, and David Mermelstein
- 45. "The 2008 Financial Crisis and Potential Output in Asia: Impact and Policy Implications" by Cyn-Young Park, Ruperto Majuca, and Josef Yap

- 46. "Do Hub-and-Spoke Free Trade Agreements Increase Trade? A Panel Data Analysis" by Jung Hur, Joseph Alba, and Donghyun Park
- 47. "Does a Leapfrogging Growth Strategy Raise Growth Rate? Some International Evidence" by Zhi Wang, Shang-Jin Wei, and Anna Wong
- 48. "Crises in Asia: Recovery and Policy Responses" by Kiseok Hong and Hsiao Chink Tang
- 49. "A New Multi-Dimensional Framework for Analyzing Regional Integration: Regional Integration Evaluation (RIE) Methodology" by Donghyun Park and Mario Arturo Ruiz Estrada
- 50. "Regional Surveillance for East Asia: How Can It Be Designed to Complement Global Surveillance?" by Shinji Takagi
- 51. "Poverty Impacts of Government Expenditure from Natural Resource Revenues" by Peter Warr, Jayant Menon, and Arief Anshory Yusuf
- 52. "Methods for Ex Ante Economic Evaluation of Free Trade Agreements" by David Cheong
- 53. "The Role of Membership Rules in Regional Organizations" by Judith Kelley
- 54. "The Political Economy of Regional Cooperation in South Asia" by V.V. Desai
- 55. "Trade Facilitation Measures under Free Trade Agreements: Are They Discriminatory against Non-Members?" by Shintaro Hamanaka, Aiken Tafgar, and Dorothea Lazaro
- 56. "Production Networks and Trade Patterns in East Asia: Regionalization or Globalization?" by Prema-chandra Athukorala
- 57. "Global Financial Regulatory Reforms: Implications for Developing Asia" by Douglas W. Arner and Cyn-Young Park
- 58. "Asia's Contribution to Global Rebalancing" by Charles Adams, Hoe Yun Jeong, and Cyn-Young Park
- 59. "Methods for Ex Post Economic Evaluation of Free Trade Agreements" by David Cheong
- 60. "Responding to the Global Financial and Economic Crisis: Meeting the Challenges in Asia" by Douglas W. Arner and Lotte Schou-Zibell
- 61. "Shaping New Regionalism in the Pacific Islands: Back to the Future?" by Satish Chand
- 62. "Organizing the Wider East Asia Region" by Christopher M. Dent

- 63. "Labour and Grassroots Civic Interests In Regional Institutions" by Helen E.S. Nesadurai
- 64. "Institutional Design of Regional Integration: Balancing Delegation and Representation" by Simon Hix
- 65. "Regional Judicial Institutions and Economic Cooperation: Lessons for Asia?" by Erik Voeten
- 66. "The Awakening Chinese Economy: Macro and Terms of. Trade Impacts on 10 Major Asia-Pacific Countries" by Yin Hua Mai, Philip Adams, Peter Dixon, and Jayant Menon
- 67. "Institutional Parameters of a Region-Wide Economic Agreement in Asia: Examination of Trans-Pacific Partnership and ASEAN+α Free Trade Agreement Approaches" by Shintaro Hamanaka
- 68. "Evolving Asian Power Balances and Alternate Conceptions for Building Regional Institutions" by Yong Wang
- 69. "ASEAN Economic Integration: Features, Fulfillments, Failures, and the Future" by Hal Hill and Jayant Menon
- 70. "Changing Impact of Fiscal Policy on Selected ASEAN Countries" by Hsiao Chink Tang, Philip Liu, and Eddie C. Cheung
- 71. "The Organizational Architecture of the Asia-Pacific: Insights from the New Institutionalism" by Stephan Haggard
- 72. "The Impact of Monetary Policy on Financial Markets in Small Open Economies: More or Less Effective During the Global Financial Crisis?" by Steven Pennings, Arief Ramayandi, and Hsiao Chink Tang
- 73. "What do Asian Countries Want the Seat at the High Table for? G20 as a New Global Economic Governance Forum and the Role of Asia" by Yoon Je Cho
- 74. "Asia's Strategic Participation in the Group of 20 for Global Economic Governance Reform: From the Perspective of International Trade" by Taeho Bark and Moonsung Kang
- 75. "ASEAN's Free Trade Agreements with the People's Republic of China, Japan, and the Republic of Korea: A Qualitative and Quantitative Analysis" by Gemma Estrada, Donghyun Park, Innwon Park, and Soonchan Park
- 76. "ASEAN-5 Macroeconomic Forecasting Using a GVAR Model" by Fei Han and Thiam Hee Ng

- 77. "Early Warning Systems in the Republic of Korea: Experiences, Lessons, and Future Steps" by Hyungmin Jung and Hoe Yun Jeong
- 78. "Trade and Investment in the Greater Mekong Subregion: Remaining Challenges and the Unfinished Policy Agenda" by Jayant Menon and Anna Cassandra Melendez
- 79. "Financial Integration in Emerging Asia: Challenges and Prospects" by Cyn-Young Park and Jong-Wha Lee
- 80. "Sequencing Regionalism: Theory, European Practice, and Lessons for Asia" by Richard E. Baldwin
- 81. "Economic Crises and Institutions for Regional Economic Cooperation" by C. Randall Henning
- 82. "Asian Regional Institutions and the Possibilities for Socializing the Behavior of States" by Amitav Acharya
- 83. "The People's Republic of China and India: Commercial Policies in the Giants" by Ganeshan Wignaraja
- 84. "What Drives Different Types of Capital Flows and Their Volatilities" by Rogelio Mercado and Cyn-Young Park
- 85. "Institution Building for African Regionalism" by Gilbert M. Khadiagala
- 86. "Impediments to Growth of the Garment and Food Industries in Cambodia: Exploring Potential Benefits of the ASEAN-PRC FTA" by Vannarith Chheang and Shintaro Hamanaka

*These papers can be downloaded from (ARIC) http://aric.adb.org/reipapers/ or (ADB) http://adb.org/Economics/publications.asp?fs=fm_9:999

The Role of the People's Republic of China in International Fragmentation and Production Networks: An Empirical Investigation

The central objective of this paper is to provide a complete description of the People's Republic of China's (PRC) parts and components trade. In order to do so, we systematically separate parts and components from final goods in the PRC's total trade flows in our empirical analysis. Our analysis indicates that the PRC is a major global production center not only of final goods but increasingly also of parts and components.

About the Asian Development Bank

ADB's vision is an Asia and Pacific region free of poverty. Its mission is to help its developing member countries reduce poverty and improve the quality of life of their people. Despite the region's many successes, it remains home to two-thirds of the world's poor: 1.8 billion people who live on less than \$2 a day, with 903 million struggling on less than \$1.25 a day. ADB is committed to reducing poverty through inclusive economic growth, environmentally sustainable growth, and regional integration.

Based in Manila, ADB is owned by 67 members, including 48 from the region. Its main instruments for helping its developing member countries are policy dialogue, loans, equity investments, guarantees, grants, and technical assistance.

Asian Development Bank 6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines www.adb.org/poverty Publication Stock No. WPS113969